Author’s response to reviews

Title: Clinical and radiological dissociation of anti-TNF plus MTX treatment in early rheumatoid arthritis in routine care. Results from the ABRAB study

Authors:

Péter Juhász (juhasz.peter@orfi.hu)
Ádám Mester (mester_ad@yahoo.com)
Júlia Bíró (biro_julia2006@yahoo.com)
Gábor Héjj (hejj.gabor@orfi.hu)
Gyula Poór (poor.gyula@orfi.hu)

Version: 3
Date: 18 June 2014

Author’s response to reviews: see over
Dear Editor,

Thank you for the comments on our article.

Sorry for the late reply, but some technical problems could occurred: no emails were received from the journal till this time, except for the last one sent on 2014.06.17.

We promptly made the appropriate corrections to fit for the journals design and uploaded the revised manuscript:

1. For fractions commas were replaced by decimals in the text and tables
2. The full name of the ethics committee was placed in the Methods section highlighted with bold letters.

Some updates were also made:
1. We put an additional affiliation for the first author: „School of PH.D. studies, Semmelweis University, Budapest, Hungary”

2. The corresponding authors affiliation was also corrected: „3rd Department of Internal Medicine, Chair of Rheumatology, Semmelweis University, Budapest, Hungary”

3. The third author suggested to change her first name from „Júlia” to „Anna-Julianna”

Please let us know if any additional corrections should be made on our article.

Yours sincerely,

prof. Gyula Poór